

Dear Pear-reviewer,

Thank you for your positive feedback and the constructive report.

Comments:

This editorial is well-written and concise. The author commented on case reports of some rare pancreatic lesions, and emphasized the importance of a comprehensive diagnostic approach by integrating medical history, cross-sectional imaging, and endoscopic ultrasound-guided tissue sampling. One minor correction is suggested. Please cite the reference (4. Liu YJ et al [to fill full citation]) in appropriate format.

Response:

In the revised version of the manuscript, I've added the full reference for Liu YJ et al, in the appropriate format.

Kind regards,

Corresponding author